Author's response to reviews

Title: Effect of an Electronic Nicotine Delivery Device (e-Cigarette) on Smoking Cessation and Reduction: A Prospective 6-Month Pilot Study

Authors:

Riccardo Polosa (polosa@unict.it)
Pasquale Caponnetto (p.caponnetto@unict.it)
Jaymin B Morjaria (J.B.Morjaria@soton.ac.uk)
Gabriella Papale (gabriellapapale@virgilio.it)
Davide Campagna (davidecampagna83@gmail.com)
Cristina Russo (kristina_russo@yahoo.com)

Version: 8 Date: 25 August 2011

Author's response to reviews:

Dear Sir,

Re.: Further formatting changes due for MS 1187998564873930 - Effect of an Electronic Nicotine Delivery Device (e-Cigarette) on Smoking Reduction and Cessation: A Prospective 6-Month Pilot Study

As per your request, we have now included the following final amendments:
Title and Abstract have been made identical to the ms in the submission system;
All additional files have been removed from the submission system

Yours sincerely, Riccardo Polosa